Morals and Mores: Experimental Evidence on Equity and Equality

By JAMES KONOW, TATSUYOSHI SAIJO AND KENJU AKAI*

Abstract

What rule is fair? This experimental study considers *equality* and *equity* (i.e., allocations that are proportional to individual contributions). Impersonal third parties, or *spectators*, favor equity. Distributive preferences move progressively toward equality, however, with the introduction of personal factors, such as sharing stakes with another (i.e., being a *stakeholder*) and lifting anonymity conditions. These findings are remarkably robust with respect to a wide range of non-ethics variables that almost never matter, including nationality, culture, race, income, and gender, and have important implications for the need to distinguish social preferences in descriptive analysis from those in prescriptive research and policy.

JEL classification: C91; D03; D63

*Corresponding author: Konow: Department of Economics, Loyola Marymount University, One LMU Drive, Suite 4200, Los Angeles, CA 90045-2659, USA (jkonow@lmu.edu); Saijo: Institute of Social and Economic Research, Osaka University, 6-1 Mihogaoka, Ibaraki, Osaka 567-0047, Japan (saijo@iser.osaka-u.ac.jp); Akai: Institute of Social and Economic Research, Osaka University, 6-1 Mihogaoka, Ibaraki, Osaka 567-0047, Japan (k-akai@iser.osaka-u.ac.jp).

We acknowledge the helpful comments and suggestions of Alexander Cappelen, Rachel Croson, Martin Dufwenberg, Robert Frank, Håkan Holm, Olof Johansson-Stenman, Stephan Kroll, David K. Levine, John T. Scott, Bertil Tungodden, and participants at meetings of the Asia Pacific Economic Science Association, Society for Social Choice and Welfare and the International Society for Justice Research, and at seminars at the National University of Singapore, Norwegian School of Economics and Business Administration (NHH), Notre Dame University, St. Andrews University, University of Gothenburg, Washington University in St. Louis, and UC Davis. We also thank Keiko Aoki, Joseph Earley, Andrew Healy, Kento Onoshiro and Kohei Ueda for their assistance with the experiment. Saijo acknowledges financial support from the Japanese Ministry of Education, Sports, Science and Technology. "It is argued that whoever does the best he can deserves equally well, and ought not in justice to be put in a position of inferiority ... On the contrary side it is contended that society receives more from the more efficient laborer; ... that, if he is only to receive as much as others, he can only be justly required to produce as much" – John Stuart Mill, 1861.

I. Introduction

What allocation rule is fair? Countless field and experimental studies have, by now, made a compelling case that fairness is an important economic force, including in labor markets, product markets, income and wealth redistribution policies, and government regulation. Nevertheless, fundamental questions that Mill raised about fairness persist almost one and one half centuries later. On the one hand, equality preferences are consistent with observed intra-firm salary compression, e.g., Frank (2004), and with the high incidence of equal splits in economics experiments, which has informed leading models of social preferences that integrate equality as an argument, e.g., Bolton and Ockenfels (2000), Charness and Rabin (2002), and Fehr and Schmidt (1999). As Crawford, Gneezy and Rottenstreich (2008) demonstrate, equal payoffs facilitate coordination, and equality has been central to much normative work, as with Rawls (1971) and Walzer (1983). On the other hand, labor markets are characterized by increasing wage inequality, which Lemieux, MacLeod and Parent (2009) trace in part to productivity differences. Moreover, in practice, most voters do not support the kind of radical redistribution equality implies, even in the most egalitarian societies, and very few people endorse equality even as a hypothetical goal: Kluegel and Smith (1986) report only 3% of survey respondents support equality or near equality of income, whereas 83% agree with paying more to more productive workers. The standard for inequality in most justice research is proportionality of rewards to contributions, which we will call "equity" (in keeping with the usage in most of the social sciences and as a way to differentiate it from equality). Equity figures importantly in philosophy, e.g., in the writings of Aristotle (1925) and Corlett (2003), and empirical studies substantiate its economic importance, e.g., Gächter and Riedl (2005), Güth (1994), Konow (2000) and Selten (1978). Equity and equality imply very large and important differences in the allocation of social and economic resources, but no consensus has yet emerged about what rule is fair, or, if multiple rules are fair, under what circumstances each rule applies.

This paper reports the results of an experiment that explores possible determinants of preferences for equity and equality. The experiment is a variation on the dictator game, in which

matched subjects first perform a task that generates earnings, and then selected subjects unilaterally decide the allocation of earnings among the group. The chief question we address is whether the relative importance of equity and equality depends systematically on personal factors, i.e., how personal the relationship is between individuals.¹ At the one extreme, we have impersonal third party dictators, or impartial spectators, who are paid a fixed fee to allocate earnings between other subjects. In other treatments, personal factors are interjected, e.g., dictators are stakeholders paired with recipients, identities are revealed, and decisions are made by teams. Research elsewhere has shown that even minor personal factors can affect allocations, e.g., Charness and Gneezy (2008) find that dictators are significantly more generous when they know the family names of their otherwise anonymous counterparts. Our question is whether personal considerations affect choices between equity and equality. Such a distinction, if found, has important consequences for theory and policy. Impersonal fairness preferences apply when personal stakes are of little or no significance. For instance, impersonal, and hence impartial, views are relevant for normative economic analysis, for third parties in experiments (e.g., Fehr and Fischbacher, 2004), and for much voting behavior as suggested, for example, by the relatively frequent incidence of affluent liberal voters and less wealthy conservative ones. Personal fairness preferences, on the other hand, arise in a host of different contexts, including with the compensation of co-workers at a firm, the distribution of assets among the beneficiaries of an estate, and the behavior between fellow stakeholders in experiments.

This study examines the propensity to act on equity or equality in relation not only to personal factors but also to the widest set of variables ever, to our knowledge. Of the five categories of variables that Camerer (2003) identifies in connection with social preference experiments, we consider ones in his four top categories, including *methodological* variables (stakes) and *demographic* variables (gender, race, age, income, etc.). Moreover, we include *cultural* and *structural* variables, which Camerer cites as having the most significant and robust effects. Regarding culture, this experiment was conducted in the US and Japan, and subjects were also evaluated using a two dimensional individualism-collectivism survey instrument.

¹ In a related study, D'Exelle and Riedl (2008) find that personal networks impact the level of giving. Our focus is, instead, on how personal relations affect the relative importance of the two main contenders for fairness rules in most justice research. The meaning of *personal* coincides here with the Webster dictionary definitions of "done without intervention of another," and "carried on between individuals directly."

Structural variables, though, are the chief ones addressed in this study, consistent with Camerer's view that they "are the most useful to study because they connect simple games to richer economic structures ... and also provide the most direct clues to the psychology underlying social preference" (pg. 75). Structural variables include here entitlement, identity and anonymity.

Entitlement is the core element in equity. In the laboratory, entitlement is typically conveyed through tasks completed by subjects, which usually produce significant effects, e.g., on bargaining in Ball, Eckel, Grossman and Zame (2001) and on public good contributions in Kroll, Cherry and Shogren (2007). Studies using entitlements indicate that relative preferences for equity or equality vary across individuals, e.g., Cappelen, Hole, Sørensen and Tungodden (2007) and Frohlich, Oppenheimer and Kurki (2004), and depend on the context, e.g., Gächter and Riedl (2006) and Konow (2000). The current study builds on and unifies these contributions by identifying general forces that impact preferences for equity and equality while revealing heterogeneity in individual responsiveness to these general forces.

Studies of possible cultural differences in social preferences have come to differing conclusions. The seminal paper of Roth et al. (1991) found that ultimatum game offers are higher in the US and Slovenia than in Japan and Israel. On the other hand, Okada and Riedl (1999) discover no significant differences between Austrian and Japanese subjects in a variation of the ultimatum game, and Brandts, Saijo and Schram (2004) also find no significant differences in public goods contributions across subjects in the US, Japan, the Netherlands and Spain. Even when behavior is the same across countries, though, Holm and Danielson (2005) argue that the underlying preferences can differ, as suggested by combined consideration of behavioral relationships and responses to survey questions. The current study similarly employs both behavioral and self-report measures to explore culture, and its results suggest an explanation for the similarities and differences in cross-cultural studies.

The laboratory method, which we employ, has both strengths and weaknesses for understanding social preferences. In light of such considerations, one suggestion of Levitt and List is "to 'nest' laboratory experiments within one another" so as "to 'net out' laboratory effects and thus reveal more about deep structural parameters" (2007, pg. 170). The design of this experiment, therefore, incorporates multiple treatments that incrementally adjust certain parameters in this manner. Indeed, our findings demonstrate the importance of this approach by indicating that actualized preferences can vary between equity and equality depending on

contextual elements found in the real world. These design features build, in part, on the relationship between social preferences and social distance that has been found not only among Western (e.g., Hoffman, McCabe and Smith, 1996) but also Asian subjects (e.g., Buchan and Croson, 2004). Moreover, we study the effects on allocation decisions of both stakeholder and spectator choices as in Konow (2000), of knowing the identity of one's counterparts (e.g., Bohnet and Frey, 1999, and Rege and Telle, 2004) and being able to communicate (Xiao and Houser, 2007), of varying the size of stakes (Johansson-Stenman, Mahmud and Martinsson, 2005), and of making individual versus team decisions (e.g., Cason and Mui, 1997).

The results show that personal factors influence the choice of equity and equality. Spectators display an exclusive preference for equity that is strikingly consistent across cultural and demographic boundaries. Nevertheless, even the weakest possible personal factor, namely being a stakeholder paired anonymously with another person, suffices to cause a significant shift toward equality. Personalizing the relationship additionally by lifting anonymity produces a further shift toward equality. Indeed, non-anonymous stakeholders equalize completely, controlling for the size of stakes. Holding constant personal factors, however, larger stakes produce more proportionate allocations. Together, these results are consistent with an impersonal preference for equity combined with a preference for equality that depends monotonically on the strength of personal considerations. In addition, average allocations do not usually differ significantly from average contributions to earnings, consistent with List's (2007) finding that tasks strengthen the sense of entitlement. Exceptions to this include a self-interested bias among anonymous big spenders and anonymous US (but not Japanese) stakeholders acting both individually and in groups. In addition, this study contributes to research on teams and finds that, although individuals are less selfish than groups, selfish groups produce individuals who are even more selfish than their groups. Other variables rarely matter, including age, income, work hours, race, gender, and cultural measures of individualism and collectivism.

Collectively, we interpret the evidence of this study as suggesting that people share a common set of social preferences and a common set of factors that impact those preferences but that there are individual, and sometimes national, differences in the weights placed on self-interest versus social preferences or on one type of social preference versus another (here equity vs. equality). We believe that this is not only the most persuasive approach to organizing the results of this study but that this interpretation can also reconcile ostensibly conflicting evidence

from many studies of social preferences. Indeed, we conclude this paper by proposing that equity and equality are but one example (albeit perhaps the most important example) of a more general distinction between two types of moral preferences, which we call *morals* and *mores*. Morals refer to the moral preferences of an impersonal or impartial third party, whereas mores are the moral preferences activated by personal considerations. In the context of these fairness rules, morals correspond to equity and mores to equality. In conclusion, we suggest that this framework can explain additional puzzles, e.g., in Crawford et al. (2008), and relate it to important recent research on identity and moral norms, e.g., Akerlof and Kranton (2000), Brekke, Kverndokk and Nyborg (2003), Ellingsen and Johannesson (2008) and McLeish and Oxoby (2007).

This paper is organized as follows. Section II reviews in greater depth the potential causes of equity and equality. Section III details the experimental design and procedures. Section IV presents and analyzes the results, and Section V contains the discussion and conclusions.

II. Equity versus Equality

Equality has a prominent place in justice research; indeed, some people equate equity with equality. Here, however, we are referring not to equal opportunity, equal rights or equal ratios, but rather to equal outcomes, i.e., *egalitarianism*. Equity, by contrast, refers in this paper to the *accountability principle* (Konow, 2000), which states that fair rewards are in proportion to the contributions that individuals control (e.g., effort and other choices). As Alesina and Angeletos (2005) show, beliefs about fairness in this vein have widespread and important implications for redistribution policies. Whereas it seems likely that equity is typically chosen simply because of a preference for it, one can think of various other explanations for why people might choose equality. One possibility is that equality occurs as a *special case* of a more general principle that otherwise generates inequality, e.g., according to the accountability principle, fair rewards are equal if the contributions people control are equal. Another reason for equality is the *ceteris paribus assumption*: when information about relevant differences is either unavailable or unreliable, people usually assume away any such differences. Evidence in favor of these two explanations for equality is reviewed in Konow (2003).

The current study examines a number of additional factors that might cause or influence the choice of equality. First among these are *structural variables*, as previously mentioned. These include the entitlement, which is implemented with a task, and various treatment variables related to personal factors, such as personal stakes, anonymity, and group (or team) decisions.

We supplement the behavioral measure of allocation decisions with self-reported evidence on motives: the study elicits subject responses about the perceived control over performance in the experimental task (known as the "locus of control" in psychology) that is relevant to the accountability principle, and about subjects' general views on the importance of the potentially competing distributive goals of equality, need and efficiency.

Second, *demographic variables* include measures of parents' income, expenditures, age, work hours and earnings. For non-anonymous decisions, we are also able to explore possible effects when the gender or race of one's counterpart is known. As Croson and Gneezy (forthcoming) report, experimental evidence on gender and social preferences is mixed. For example, Eckel and Grossman (2001) find that women are more generous and exhibit greater solidarity with one another than men, whereas Ben-Ner, Kong and Putterman (2004) find the opposite. In Dufwenberg and Muren (2006), what is mostly relevant is the gender of the recipient: women receive more than men. The evidence on race and social preferences is similarly mixed: blacks are more generous in the ultimatum study of Eckel and Grossman (2001) whereas race does not significantly affect average dictator giving in Fong and Luttmer (2009).

Third, we explore possible effects of *nationality and culture* on distributive preferences among US and Japanese subjects.² One criticism of multi-national economics experiments is that they fail to distinguish nationality from culture, whereas our data allow us to do so at the individual level (something that might be particularly useful when dealing with a culturally diverse country like the US). Previous multi-national economics experiments have examined cooperation, trust and reciprocity. To our knowledge, this study is unique among such inquiries, however, in isolating distributive preferences both from intentional preferences and from self-interest in a cross-national context.

National differences in justice, if present, can be interpreted in a number of ways. One

² There are four reasons we chose the US and Japan for this study. First, they are the two largest economies in the world and represent, therefore, a substantial fraction of world economic activity. Second, the largest volume of experimental economics has been conducted in the US, and Japan is probably the other country that has most frequently been used in international comparisons. This assists efforts to generalize findings by facilitating connections between this study and others. Third, the US and Japan are often said to represent two sides of the most commonly cited cultural divide, viz., Eastern collectivism versus Western individualism. Fourth, our chief interest is in preferences for equality versus inequality, and much cross-national research places these two countries on opposite sides of this issue.

view is that justice is simply not as strong a value in Eastern as in Western countries (Greenberg 2001), which seems consistent with findings of Mattila and Patterson (2004). The most common interpretation, however, is probably that Westerners favor equity whereas Easterners prefer equality, in line with the results of Mann, Radford and Kanagawa (1985) and Kashima et al. (1988). Traditionally, Japan experienced one of the lowest levels of income inequality in world, which Tachibanaki (2005) attributes to the strong "egalitarian principle prevalent among the Japanese" (pg.110). Although inequality has increased substantially in recent decades, this does not seem to reflect a change in values. Rather, Ohtake and Saito (1998) point out that the income distribution by age has remained relatively constant, and they attribute the increasing overall inequality to Japan's rapidly aging population. Yet another explanation for cross-national differences in behavior is that, independent of whether or not people hold to the same justice concepts, they differ in their inclination to act on them. Cason, Saijo and Yamato (2002), for example, find that Japanese subjects.³ For this reason, we find it helpful in this study to distinguish the choice of fairness rules from the degree of rule compliance.

Fourth, a potentially important variable is the size of the *stakes*. Summarizing a set of experiments, Güth (1988) concludes that the basic rule of distributive justice is proportionality but observes that smaller stakes are often associated with equality. Although one can consider stakes a methodological variable, we propose that its effects are not merely artifactual. Instead, this provides an additional means to test the hypothesized trade-off between equity and equality: for given personal factors, higher stakes should increase the importance of equity. This also mirrors questions outside the laboratory: one common real world example is the tendency to "split the tab" in restaurants when the orders of diners do not differ greatly, thereby avoiding various costs, notably goodwill, of a more exact reckoning. A more economically significant case is the reluctance by employers (and even employees) to support fully compensating productivity differences within a firm in the interests of promoting collegiality, whereas such differences are more acceptable across industries or for national income policies. We are unaware of any previous empirical test of this trade-off.

³ Their study also indicates that US and Japanese subject pool effects are mostly due to national differences rather than institutional differences: their experiment was conducted at two universities in the US and two in Japan, and they found significant between-country differences but comparatively minor within-country differences.

III. The Experiment

A. Design

There are two phases to the experiment. In the production phase, twelve subjects in room X and twelve in room Y perform a task that generates earnings (described in section *III.B*). In the subsequent allocation phase, subjects are matched and their joint earnings are allocated among them dictator style. The three treatments of the experiment, which are summarized in Figure I, differ with respect to who makes the allocation decision (i.e., who is dictator), how subjects are matched, and the level of anonymity. In the Spectator treatment, subjects in Rooms X and Y are anonymously matched in pairs. Usually, subjects differ considerably in their performance on the task and, therefore, in the earnings that can be attributed to them individually. The matching protocol takes advantage of this to produce the widest range of productivity differences within pairs by pairing, for each session, the most productive X subject with the least productive Y subject, etc. Then a spectator, or third party, makes the eponymous dictator decision. Specifically, twelve spectators are located in a third room, called Z, and each is individually and anonymously matched with a single X/Y pair. Spectators are paid a fixed fee, unrelated to their allocation decisions, to distribute the earnings generated by their X and Y counterparts between them.

Treatment	Decision	Dictators \rightarrow Recipients	Anonymity
Spectator	Spectator	$Z \rightarrow X, Y$	Anonymous
Anonymous Stakeholder	Anonymous Stakeholder	$X \rightarrow X, Y$	Anonymous
Group	1. Group	$X (XA/XB) \rightarrow \begin{array}{c} X (XA/XB), \\ Y (YA/YB) \end{array}$	Anonymous between X and Y Known between A and B
Stakeholder	2. Known Stakeholder	$\begin{array}{ccc} XA & \rightarrow & XA, XB \\ YA & \rightarrow & YA, YB \end{array}$	Known

FIGURE I. Experimental Design

The Anonymous Stakeholder treatment is closer to the standard version of the dictator game. Subjects in rooms X and Y (again twelve each) first generate earnings and are then anonymously matched into pairs as in the Spectator treatment, but there is no third party Z. Instead, X subjects are arbitrarily chosen to allocate the earnings generated by their pair between themselves and their Y counterparts. Each X subject does this individually and anonymously for his or her pair. Thus, the only difference between the Anonymous Stakeholder and Spectator decisions is whether or not the dictator is also a recipient, i.e., a party to the earnings being distributed, the weakest personalization of a relationship where material stakes are involved.

The Group Stakeholder treatment begins with the same production phase as the other treatments, but the allocation phase involves two stages and a different matching protocol. Instead of pairs, subjects are initially matched into quadruples consisting of two X subjects, called XA and XB who form Group X, and two Y subjects, called YA and YB who form Group Y, and the earnings of all four subjects are pooled. The first allocation involves a Group decision. XA and XB subjects are re-seated to meet face-to-face, are informed of the total (but not individual or group) production and earnings of their quadruple, and jointly choose how much to take for their own Group X and how much to give to Group Y. YA and YB also meet face-to-face and are informed of the Group X decision. Then, A and B subjects in both rooms return to their original seats. Although A and B subjects in both rooms meet one other, X and Y groups never meet and remain anonymous to one another.

The second decision in the Group Stakeholder treatment is an individual one. Subject XA is arbitrarily chosen to allocate the Group X earnings they just selected for themselves between himself and his XB counterpart. Similarly, the YA subject is arbitrarily chosen to distribute the earnings Group X gave to them in the earlier decision between herself and her YB counterpart. These decisions are all made individually, but they are not anonymous: A and B subjects have met and know one another's identity. This Known Stakeholder decision is similar to the Anonymous Stakeholder decision in that the dictators are individual stakeholders but differs in that they are known by and known to the recipients in this case. Thus, comparison of these two decisions identifies the effect on individual stakeholder allocations of lifting anonymity.

The matching mechanism in the Group Stakeholder treatment is a bit more involved. In each room, A and B subjects are separately matched so as to maximize productivity differences, i.e., the most productive A with the least productive B, etc., analogous to the subject X and Y pairings in the other treatments. Then these X groups (each consisting of an XA and XB) and Y groups (consisting of YA and YB) are matched to form quadruples, specifically, the most productive X group is matched with the most productive Y group, the second most productive X group with the second to most productive Y group, etc. This results in relatively large average differences in productivity between A and B subjects but relatively small average productivity differences between X groups and Y groups. This allows us to focus on aspects of self-interest in

the first Group decision and on equity in the second Known Stakeholder decision. Specifically, this treatment also relates to the research on team decision making. Cason and Mui (1997) find that individuals act more selfishly than groups, although most other work in this area, including Luhan, Kocher and Sutter (2009), indicate the opposite. This study tends to corroborate the latter and adds to the evidence on the effects of groups on individuals, as explained in section IV.⁴

B. Procedures

A total of 432 subjects participated in this experiment: 144 in the Spectator treatment, 96 in the Anonymous Stakeholder treatment and 192 in the Group Stakeholder treatment. Each of these totals consisted of equal numbers of subjects from the undergraduate campuses of universities in Los Angeles, California and Osaka, Japan. Subjects were invited by campus wide emails and flyers posted around campus to sign up at designated websites. All sessions had twelve subjects per room, and the Spectator treatment was conducted with three rooms (X, Y and Z), or 36 subjects total, per session, whereas the Anonymous Stakeholder and Group Stakeholder treatments each involved two rooms (X and Y), or 24 subjects total, per session. All subjects initially showed up at a single location to register and receive their show up fees before being assigned to their rooms in order to dispel doubts about the existence of counterparts in other rooms (see Frohlich, Oppenheimer and Moore, 2001, for evidence on the effects of such doubts).

After random assignment to separate rooms, subjects are told that there are two phases of the experiment and then given more specific instructions for the first phase.⁵ In the production

⁴ There are two additional reasons for this sequence of decisions and for matching subjects so as to maximize A/B differences and to minimize Group X/Group Y differences. First, our aim is to investigate incrementally the impact of personal relations on the choice of equality or equity, and the latter requires significant differences in productivity between subjects. In this sense, the next increment from the Anonymous Stakeholder decision in the Group Stakeholder treatments is the second Known Stakeholder decision, not the first Group decision. Moreover, given the statistical pattern of subject productivity, one must choose to maximize differences between either the one or the other: when disparate A and B subjects are paired, the A/B (i.e., the group) totals do not vary much. Indeed, many group totals varied only by one or two units (in fact, the A/B pairs in one group all produced exactly the same total). The reason information about group totals is not provided to X groups is because group totals do not vary much anyway and the focus, therefore, of the Group decision is on self-interest versus equality. Second, we are also interested in looking at the effect of pie size (i.e., the amount of earnings to be distributed), and a natural way to create variation without losing A/B differences is to have the Group decision precede the Known Stakeholder decision: one can expect considerable variation in how much X groups leave to Y groups, although usually less than one-half, and this expectation was borne out. ⁵ As in previous experiments of this kind, subjects were initially only informed in general terms

phase, subjects prepare letters for mailing, a task that has been previously shown to produce a clear and strong sense of entitlement (Konow, 2000). Each letter correctly prepared in the six minutes allotted generates 100 points, which is also stated in local currency (i.e., US dollar or Japanese yen). After the task is complete and the letters counted, the pooling of earnings for their particular treatment is explained, the dictators are identified for the first time (although they are merely identified as subjects X or Z), and dictators are given five minutes to allocate points between subjects. The Group Stakeholder treatment is a bit more involved. First, the X groups are given five minutes to allocate the total of the quadruple between themselves and the Y group.⁶ Then, the Known Stakeholder allocations take place: the A subjects in each room are given five minutes to allocate the group X (or Y) totals between themselves and B subjects. Finally, all subjects complete a questionnaire, viz., the Singelis et al. (1995) four scale individualism-collectivism instrument, whereby dictators also answer a question asking why they allocated as they did, after which subjects are paid privately and permitted to leave.

Although the primary focus of this study is not culture, we controlled for country conditions in a number of ways. Subjects at both venues were recruited in the same manner described above from across a wide range of undergraduate disciplines. The two universities are located in comparably sized metropolitan areas, which in both cases are the second largest in their respective countries. Subjects in both locations were screened to exclude non-citizens and were recruited and assigned to rooms so as to maintain an approximately equal number of men and women, where possible, particularly in the dictator rooms. We controlled for purchasing power parity using OECD conversion rates. On this approximate basis, show-up fees were \$5 in the US and 750 yen in Japan, and each letter in the US earned \$1 (1 point = 1 cent) and in Japan earned 150 yen (1 point = 1.5 yen). The instructions were written in English, translated into Japanese and then back-translated by a separate translator into English to check for consistency. The first author was present at both locations to verify that the recruitment, procedures, and even

about allocation procedures in order to avoid any effect on effort and productivity in the first stage that might subsequently impact allocations and confound inferences about distributive motives. The experiment was partially, but not completely, computerized for logistical reasons, viz., one of the labs did not have separate computerized rooms, but the use of pencil and paper for allocation decisions probably also helped reinforce that dictator decisions were anonymous to recipients. For practical reasons, though, decisions were not blind to the experimenter. ⁶ The X groups are told that, if at the end of the five minutes they fail to agree, one of the X subjects will be randomly chosen to decide, although it never came to that.

physical set-up were equivalent. For language reasons and in order not arouse suspicion, the experiment was conducted solely by Americans in the US and Japanese in Japan, and the lead experimenter in the dictator rooms (except for the YA decisions) was always the same person (Konow in the US and Akai in Japan). The experimental protocol is contained in the Appendix.

IV. Results and Analysis

Participants in this experiment earned, on average, \$18.14 in the US and 2121 yen in Japan (or about US\$20 at contemporaneous exchange rates or about \$15 in US purchasing power) including show up fees for sessions that lasted, on average, about 50 minutes. After receiving their payments, 99% of American subjects and 94% of Japanese subjects responded that they would be willing to participate in other economics experiments. In section IV.A, we review and analyze the results on the relationship of equity and equality to the structural variables of interest, i.e., entitlement, personal factors and stakes. Section IV.B addresses other potentially related considerations, including demographic variables and nationality/culture.

A. Equity, Equality and Structural Variables

Entitlement, stakeholding and anonymity

SUMMARY OF MEAN RESULTS					
	Spectator	Anonymous Stakeholder	Known Stakeholder	Group Stakeholder	
Allocations Mean (Std. Err.)	0.45 (0.036)	0.57 (0.029)	0.54 (0.015)	0.59 (0.023)	
Entitlements Mean (Std. Err.)	0.45 (0.033)	0.49 (0.028)	0.52 (0.019)	0.46 (0.006)	
Differences in t-statistic p-value	n means 0.01 0.994	1.89 0.061	0.84 0.401	5.63 0.001	
No. of obs.	48	48	91	48	

TABLE I Summary of Mean Results

Notes: There are only 91 decisions by Known stakeholders (48 XA and 43 YA subjects), because five of the 48 X Groups allocated nothing to their Y Groups leaving no decisions for the corresponding five YA subjects.

Table I summarizes the mean allocations to X (or A, in the case of Known Stakeholders) as fractions of total earnings for the four allocation decisions. The entitlement is the mean

fraction of earnings produced by X or A, respectively. Two-tail t-tests of differences between mean allocations and mean entitlements demonstrate no significance difference for Spectator and Known Stakeholder allocations, whereas Anonymous Stakeholders take 8 percentage points more than their entitlement, a difference that is marginally significant. Only Group Stakeholder allocations exceed the fraction they produced at conventional levels of significance, although remember that, for practical design reasons, these dictators were informed only about the total earnings of their quadruple but not about the letters produced separately by groups or individuals. Nevertheless, their allocations also differ significantly from the equal splits that are consistent with the aforementioned ceteris paribus assumption, which is appropriate under these information conditions (t=4.04, two-tail p-value < 0.001). Thus, these results suggest variation in bias toward one set of subjects ranging from no significant bias when, at the one extreme, dictators are informed third parties in the Spectator treatment to a large and significant effect of self-interest when, at the other extreme, groups of uninformed stakeholders who are known to one another but anonymous to recipients allocate in the Group Stakeholder treatment.

TYPE OF ALLOCATION BY DECISION	
(percentage of each type)	
Distator Desigion	

TARI F II

	Dictator Decision				
Allocation		Anonymous	Known	Group	
Туре	Spectator	Stakeholder	Stakeholder	Stakeholder	
Proportional	81	57	45	NA	
Equal	19	30	49	90	
Selfish	NA	13	6	10	

Table II presents a simple but somewhat more detailed summary of the results. It categorizes each decision into one of three types, viz., proportional, equal or selfish, based on a calculation of whether it is closest in absolute terms from its respective entitlement, equality or giving X (or A in the Known Stakeholder decision) the entire pie (ties are counted one-half to each type). The first column illustrates that 81% of Spectator decisions are closest to being proportional and 19% to being equal (the selfish category is not applicable here, since the dictators in this treatment are third parties). The percentage of proportional decisions falls to 57% and the percentage of equal ones rises to 30% for Anonymous Stakeholders, while 13% of decisions are closest to the completely selfish allocation. A further trend toward equality and away from proportionality is found for Known Stakeholders, although selfish allocations fall

somewhat. Thus, these results imply a shift away from proportionality and toward equality when decision-makers are stakeholders and again when they are known to and by their recipients. Finally, we see that, although Group Stakeholders deviate, on average, significantly toward self-interest, as reported above, 90% of their allocations are still closer to equality than to taking the entire earnings. Here the proportional category is not meaningful, since these dictators receive no information about relative production. This is also the reason for concentrating on the three individual decisions in most of the following analysis, which concerns equity versus equality.



FIGURE II. Dictator Allocations

The dictator allocations of Z subjects in the Spectator treatment are illustrated in Figure IIa. The horizontal axis represents the fraction of letters produced by the X subject and the vertical axis the fraction of earnings allocated to the same X subject by the Spectator. If spectators value equality, the allocations should lie along a horizontal line at 0.5. Equity, on the other hand, calls for proportionality: fair allocations, or the *entitlement*, lie along the lighter 45 degree line where fractional allocations equal fractional contributions (ignore the dark lines for now). Apart from a few outliers, most decisions appear to be quite close to the entitlement.

Figure IIb presents the results of the Anonymous Stakeholder treatment. The points refer again to the X subjects, but the fractional earnings on the vertical axis are those chosen by the X subjects to themselves (rather than those chosen by a third party). As with Spectators, a number of these allocations equal the entitlement, but departures from this line are more prominent. The Known Stakeholder treatment in Figure IIc illustrates the dictator allocations of A subjects in rooms X and Y to themselves, whereby they are known to and by their B counterparts. The frequency of equal split allocations appears to be greater here than in the other treatments. These impressions are reinforced and further illuminated by regression analysis. The dark lines in Figure II result from the following OLS regression of fractional allocations on fractional entitlements for the three individual decisions:

(1) Allocation_i = a + b·Entitlement_i + ε_i

Equality corresponds to an intercept of one-half (a=0.5) and a slope of zero (b=0), whereas equity predicts an intercept of zero (a=0) and a slope of one (b=1). Since the dependent variable is left- and right-censored, however, we also ran two-sided Tobit regressions for these and the other regressions reported in this paper.⁷ Table III reports these regressions and the results of F-tests of the joint hypotheses for the equity and equality cases. For the Spectator decisions, there is strong support for equity and no support for equality, confirming the impressions from Table II and Figure II. Allocations in the two stakeholder decisions, however, fall between equity and equality and differ significantly from those two sets of predictions, as seen in Table III. Comparing slope coefficients to those of Spectators, dictators equalize more as Anonymous Stakeholders (t-statistic=-2.12, p<0.05) and Known Stakeholders (t-statistic=-3.97, p<0.01), indicating that personalizing the interaction has an equalizing effect. Joint tests of the hypothesis of no change in intercepts and slopes similarly indicate significant differences in Spectator allocations relative to those of Anonymous Stakeholders (F-statistic=5.34, p<.01) and Known Stakeholders (F-statistic=10.55, p<.01).

The significant slope coefficients in all three regressions demonstrate that the entitlement, and therefore equity, matters in every condition, although to differing degrees. Spectators allocate according to equity, whereas Anonymous Stakeholders respond to but are less sensitive than spectators to differences in entitlement while their allocations reflect a marginally significant influence from self-interest, as we know from Table I. Dropping anonymity in the Known Stakeholder decision reduces the roles of equity and of self-interest, according to Tables I, II and III.⁸

⁷ For the findings reported in this paper, OLS and Tobit regressions result in conclusions that are qualitatively, and even quantitatively, very close. Nevertheless, it is worthwhile establishing the robustness of the results, given the focus of this study on preferences for equality: stakeholders might allocate in a more self-interested manner resulting in disproportionate censoring of the allocations of more productive dictators and lower OLS slope coefficients, even if their fairness rule were equity. This is a problem of right censored data, which these Tobit regressions correct.

⁸ This last effect is consistent with the results of Hoffman et al. (1996), but see also Dufwenberg and Muren (2006) whose novel findings indicate the opposite.

Intercept	Spectator 0.01	Anonymous Stakeholder 0.24***	Known Stakeholder 0.30***
	(0.049)	(0.072)	(0.037)
Entitlement	0.96*** (0.097)	0.67*** (0.138)	0.47*** (0.067)
F-statistics Equity (a=0, b=1) Equality (a=0.5, b=	0.11 =0) 52.39***	7.61*** 15.67***	33.16*** 31.40***
Adjusted R^2	0.70	0.33	0.34
No. of obs.	48	48	91

 TABLE III

 TOBIT REGRESSION ANALYSIS

Notes: Standard errors are in parentheses. Adjusted R^2 are from OLS. Significant at *10%, **5%, ***1% level.

The results in Figure II suggest patterned differences in the application of fairness rules not only across treatments but also across subjects. One explanation for this is that subjects fundamentally agree on equity and equality but differ in their interpretation of the appropriateness of the rules in the context, e.g., some stakeholders believe that merely being paired with another person requires that they equalize (or equalize somewhat) whereas others believe the relationship must be more personal before relinquishing proportionality. We consider two other conjectures about individual heterogeneity related to beliefs about accountability and its importance. First, different allocation decisions might reflect differences across subjects in how they interpret accountability in the context of the experimental task: those who allocate proportionally see the task as being under the control of subjects whereas those who equalize do not. Another possibility is that subjects differ fundamentally in the importance they attach to accountability versus other distributive goals such as equality, need and efficiency.

The experimental questionnaire contains five questions (11 through 15 in the Appendix) that ask for agreement or disagreement with statements motivated by these two conjectures. Question 11 addresses accountability in the experimental task and the other questions concern general distributive preferences, viz., need in question 12, equality in 13, efficiency with equality in 14, and efficiency with equity in 15. In analysis not shown here, we add five variables to equation (1) for each of the five questions and conduct three separate regressions for the

Spectator, Anonymous Stakeholder and Known Stakeholder decisions. In all of these regressions, the entitlement continues to be positive and significant, but none of the questions is significant. There are still no significant coefficients on these questions if we interact them with the entitlement and conduct the regressions separately for each of the five questions and three decisions. These findings suggest that differences in dictator allocation behavior are not due to differences in the interpretation of accountability in the experiment or to more general differences in the distributive preferences of subjects. Indeed, an examination of the raw responses indicates broad consensus among subjects in favor of accountability and about the relative importance of other distributive principles.⁹ Instead, the results tend to support differences associated with the personal factors (i.e., personal stakes and anonymity) and how subjects interpret and respond to these factors.

Finally, remember from Table I that Group Stakeholders take a larger fraction of earnings than any other set of individual decision makers in this experiment, consistent with most previous work. Nevertheless, additional Tobit analysis (not shown here) indicates that the subsequent allocations of Known Stakeholders in Group X to themselves are significantly positively related to the fraction the X group took for itself. Luhan, Kocher and Sutter (2009) find that, after being in groups, individuals behave more selfishly than they did before, but not that they are more selfish than the groups. Here, however, the implication is that individuals who belonged to selfish groups subsequently act more selfishly than even their selfish groups. This is especially noteworthy, given that individual dictators allocated to recipients known to them, unlike similar prior experiments in which individual decisions were anonymous.

Size of stakes

As previously discussed, one reason sometimes given for equality is that people forgo proportional accounting when the stakes are not very large, in particular, when decisions are not

⁹ For question 11, 79% of subjects agree that "the performance of subjects on the task in this experiment was due mostly to things that they could control" (collapsing agree and strongly agree into one category and disagree and strongly disagree into another). For question 12, 64% agree "the basic needs of people for food, clothing and shelter differ greatly across individuals," the lowest level of consensus for any of the questions. On question 13, 86% disagree with the statement that "there should be the same income guarantee for all people." For question 14, 90% disagree that "the way for companies to be most productive and efficient is by minimizing differences in pay across workers," whereas, on question 15, 87% agree that "the way for companies to be most productive according to their productivty."

anonymous. This provides a separate test of equity as an impersonal rule and equality as a personal one: when personal factors are held constant within a given decision, higher stakes should increase the weight given to equity considerations. To test this prediction, we include the size of the Stakes in OLS and Tobit versions of the following regression:¹⁰

(2) Allocation_i = a + b·Entitlement_i + c·Stakes_i + d·Entitlement_i·Stakes_i + ε_i

If the prediction about stakes is correct, the coefficient on c should be negative and the coefficient on d positive: as stakes increase, dictators increasingly depart from equality and approach proportionality.

	IOWN BTAKEHOEDER MELOCATIONS
Variable	Parameter estimate (Std. Err.)
Intercept	0.511***
	(0.103)
Entitlement	-0.023
	(0.180)
Stakes	-0.252**
	(0.111)
Entitlement × Stakes	0.556***
	(0.196)
Adjusted R^2	0.41
Number of observations	91

 TABLE IV

 Effect of Stakes on Known Stakeholder Allocations

Notes: Tobit regressions with adjusted R^2 from OLS. Significant at *10%, **5%, ***1%.

We focus on the Known Stakeholder results, since only this case fits the story involving non-anonymous decisions, and since it is the only decision in this experiment with sufficiently large variance in stakes to test meaningfully its effect.¹¹ The Tobit regression results are

¹⁰ To ease interpretation, the Stakes variable was created as follows: the total points available to each X/Y or A/B pair, respectively, are divided by the average total points across all pairs in the Spectator, Anonymous Stakeholder and Known Stakeholder decisions. Thus, the average stakes across these decisions equal 1, while the allocation and entitlement continue to be measured as fractions of the individual stakes.

¹¹ The variance of standardized stakes in the Spectator and Anonymous Stakeholder decisions are both 0.06, whereas it is 0.27 (more than four times greater) in the Known Stakeholder

presented in Table IV and confirm all predictions of the hypothesis: larger Stakes significantly decrease the intercept, with an estimated coefficient of -0.252, and significantly increase the slope, with a parameter estimate on the interaction term of 0.556. Thus, larger stakes are associated with significantly more proportional allocations, or, put differently, dictators allocate more equally with smaller stakes.¹² Indeed, this regression provides the strongest evidence of equality: controlling for the size of stakes, allocations in this, the most personalized, treatment do not differ significantly from an intercept of 0.5 and a slope coefficient of 0 on the Entitlement.

In unreported analysis, we consider an alternate conjecture about stakes being correlated with an omitted variable that causes more equal allocations. Specifically, suppose whether or not one was previously in a position of power affects one's subsequent equality preferences. That is, YA dictators might equalize more, on average, not because of the smaller stakes they typically receive, but because of sympathy or solidarity with YB dictators who were similarly powerless in the first round. We add a dummy variable for YA Known Stakeholders and an interaction term with the entitlement to equation (2) to test this conjecture, but these terms are not significant. Thus, it appears that the size of stakes per se drives these results.

B. Other Potential Factors

This subsection considers the potential effects of other variables on the magnitude of dictator generosity and on the question of whether it is motivated by a preference for equity or equality. Although we test for a large set of possible effects, statistically significant factors turn out to be rather rare, suggesting that moral preferences dominate behavior in these cases. Nationality and Culture

We begin by comparing mean allocations and mean entitlements for the three individual decisions by country in Table V. Of the six decision/country comparisons, the only significant difference between allocations and entitlements is for the US Anonymous Stakeholder case. Thus, there is no significant evidence of bias in any decisions of Japanese subjects, and the one indication of bias, which was marginally significant in the pooled sample with Anonymous

decision, significantly larger than the former two and a much better basis for evaluating the potential impact of stakes. Indeed, in separate regressions for Spectators and Anonymous Stakeholders, neither c nor d is significant in the US, Japan and pooled samples.

¹² The results remain qualitatively the same, if we run this regression using OLS or on the US and Japanese sub-samples: the estimates of a and d are significantly positive, c significantly negative and b insignificant.

	Spectator		Anon	Anonymous Stakeholder		Known Stakeholder	
	US	Japan	US	Japan	US	Japan	
Allocations		1		1		<u> </u>	
Mean	0.40	0.50	0.57	0.57	0.56	0.53	
(Std. Err.)	(0.058)	(0.039)	(0.040)	(0.044)	(0.021)	(0.020)	
Entitlements							
Mean	0.38	0.52	0.47	0.52	0.52	0.53	
(Std. Err.)	(0.053)	(0.037)	(0.047)	(0.023)	(0.023)	(0.029)	
Differences in	n means						
t-statistic	0.20	-0.309	2.06	0.79	1.31	0.00	
p-value	0.839	0.758	0.045	0.434	0.195	0.999	
No. of obs.	24	24	24	24	44	47	

TABLE V Summary of Mean Results by Country

Stakeholders, is actually due only to self-interest on the part of US subjects.¹³

The Group Stakeholder decisions suggest a similar pattern of self-interest as the Anonymous Stakeholders. Remember that groups do not know entitlements in this decision, so we compare group allocations to equal splits. In the US, X Groups allocated, on average, a fraction of 0.64 (SE=0.036) to themselves, or an amount that differs significantly from one-half of joint earnings (t=3.93, two tail P-value < 0.001). Japanese X Groups, on the other hand, allocated an average of 0.54 (SE=0.025) to themselves, a difference from one-half that is only marginally significant (t=1.72, two tail P-value = 0.098).

¹³ One conjecture about the lower level of self-interest in Japan is that it is due to single-blind anonymity: Japanese subjects might be more inclined than US subjects to alter their decisions in order to present themselves more favorably to the experimenter. As mentioned previously, we did not see a way to conduct this experiment double blind, but any concern about this should, in any case, be allayed by a number of facts. First, the lead experimenter in the dictator room helped with registration, and whenever he recognized a subject, the subject was specifically assigned to a different room. Second, the lead Japanese experimenter was a graduate student who, in both countries, has lower social status than a professor, who was used in the US. Third, we know of no evidence that single blind procedures prompt any different response among Japanese subjects than among Western ones. For example, Okada and Riedl (1999) use the single blind method and find no significant difference, indeed, evidence elsewhere does not support any strong experimenter effect in general (e.g., Cason and Mui, 1997, and Bolton et al., 1998). Moreover, our primary interest is in equity versus equality, and if Japanese allocations are a

Moreover, our primary interest is in equity versus equality, and if Japanese allocations are a response to the experimenter, then why, for example, is the tendency of Japanese Stakeholders to equalize not also observed among Japanese dictators in the Spectator treatment?

Table VI presents Tobit regression results similar to those in Table III but decomposed into US and Japanese subjects. The results for these national sub-samples are quite similar to those for the pooled sample, according to F-tests. Spectators in both countries are striking in their adherence to equity: their allocations do not deviate significantly from proportionality but do differ from equality. Stakeholder allocations lie between equity and equality with Known Stakeholders equalizing more than Anonymous ones. Moreover, US and Japanese do not differ greatly with respect to equalization, although there are mixed results on Anonymous Stakeholders in Japan: their regression coefficients imply they equalize somewhat more but joint F-tests of intercept and slope suggest their departure from equity drops slightly in significance (p<0.10 with Tobit although p<0.05 with OLS). Overall, therefore, these findings suggest a similar pattern of distributive preferences and, in any case, provide no significant support for the claims that Japanese value equality more than Americans.

	Spectator		Anony Stakeł	Anonymous Stakeholder		Known Stakeholder	
	US	Japan	US	Japan	US	Japan	
Intercept	-0.01 (0.070)	0.04 (0.072)	0.22* (0.116)	0.23** (0.099)	0.33*** (0.068)	0.27*** (0.040)	
Entitlement	1.06*** (0.156)	0.88*** (0.132)	0.75*** (0.239)	0.66*** (0.175)	0.44*** (0.126)	0.49*** (0.071)	
F-statistics Equity Equality	0.09 27.32***	0.71 22.36***	5.27** 7.07***	2.86* 8.85***	12.33*** 10.85***	26.12*** 26.39***	
Adjusted R^2	0.69	0.65	0.28	0.37	0.22	0.49	
No. of obs.	24	24	24	24	44	47	

TABLE VI TOBIT REGRESSION RESULTS BY COUNTRY

Notes: Standard errors are in parentheses. Adjusted R^2 are from OLS. Significant at *10%, **5%, ***1% level.

Although this analysis reveals no striking differences in preferences for equity and equality between the US and Japan, one cannot rule out some cultural influence on these preferences that is orthogonal to nationality. Indeed, even when behavioral differences do emerge between subjects in different countries, there is a potential ambiguity about the source. Are they due to nation-specific attributes or to more general cultural traits, which might vary

both within and across different countries? To address these questions and to give culture its best shot, this experiment identifies culture separately from nationality. Specifically, after the allocation decision, subjects in both countries completed a questionnaire that included a measure of individualism vs. collectivism. A typical view is that Eastern societies are more collectivist (i.e., they define themselves as a group), whereas Western societies are more individualistic (i.e., they define themselves as individuals). We use the Singelis et al. (1995) instrument, which is one of the richer measures of culture and is particularly well suited to our topic. It is based on a distinction along two dimensions pertaining to in-group/out-group and equality/inequality preferences. Collectivists prefer their in-group (e.g., class or caste), whereby vertical collectivists (VC) accept in-group inequalities (e.g., India), whereas horizontal collectivists (HC) do not (e.g., Kibbutzim). Individualists do not make stark in-group/out-group distinctions, and in the vertical version (VI) individual inequality is accepted (e.g., France), whereas in horizontal individualism (HI), each person is more or less equal (e.g., Sweden). This instrument consists of four culture scales corresponding to these four types (HI, VI, HC and VC), each containing eight questions. Each scale can assume values between 8 and 72 inclusive, where higher values indicate a stronger cultural orientation in that direction.¹⁴ Singelis et al. suggest that the US and Japan have mixed cultural orientations, but that the former is mostly VI and the latter mostly VC.

Table VII summarizes the mean scores on these scales for all subjects by country. US subjects score significantly higher on all scales, save VI, where the two groups do not differ significantly. This provides no support, therefore, to the expectation of Singelis et al. that Americans are more VI or that Japanese are more VC. We test the effects of culture on allocations by adding all four scales as regressors to equation (1) in addition to interacting these scales with the entitlement. We ran three separate regressions for Spectators, Anonymous Stakeholders and Known Stakeholders in pooled samples of US and Japanese subjects (not shown). None of the culture scales is significant, except the VC scale in the regression for Known Stakeholders. In this regression, the only significant variables are the entitlement, VC and the interaction of VC with the entitlement. Specifically, the signs on these variables indicate that Known Stakeholders take more, the larger the fraction of earnings they generated, and that

¹⁴ Subjects respond to statements on a scale from 1 to 9, and each culture scale is formed by simple addition of these responses for its respective eight questions. The individual items can be found in the Appendix, whereby HI is formed from questions 1-8, VI from 9-16 with 16 reverse scored, HC from 17-24, and VC from 25-32.

those with a greater VC orientation equalize less and allocate more proportionately. This seems consistent with the notion that vertical collectivists accept more in-group inequalities. On the other hand, what is particularly noteworthy about these results is the relative unimportance otherwise of these cultural measures for allocations.

WIEAN COLTORE SCORES BT COONTRT							
					Test of Difference		
	US			Japan	in Means		
					(Ho:US=JP)		
	No of	Mean score	No of	Mean score	t-statistic		
	Obs.	(Std. Err.)	Obs.	(Std. Err.)	(P-value, two tail)		
Horizontal	216	54.26	216	43.62	15.01		
Individualism (HI)		(0.450)		(0.548)	(0.000)		
Vertical	215	43.92	215	44.27	-0.37		
Individualism (VI)		(0.744)		(0.587)	(0.713)		
Horizontal 216		58.16	215	48.81	11.52		
Collectivism (HC)	210	(0.522)	213	(0.621)	(0.000)		
Vertical	215	44.67	216	38.50	7.15		
Collectivism (VC)	213	(0.648)	210	(0.573)	(0.000)		

TABLE VIIMean Culture Scores by Country

To summarize, we find little evidence of cultural variation in allocation behavior according to the individualism-collectivism instrument used. Fairness preferences do not appear to differ between US and Japanese subjects: spectators apply equity, and stakeholders strike a balance between equity and equality, whereby lifting anonymity causes a further shift toward equality and away from equity. The only national differences we find are the significant selfinterest of US subjects, both individually and in groups, when they are anonymous stakeholders, compared with the absence of significant self-interest among any set of Japanese subjects. Demographic Variables

In analyses not reported here, we examined the effect of various demographic variables on dictator giving. First, we added the following regressors to equation (1): age, student expenditures during the school year, student earnings, parents' annual income (in seven discrete categories), work hours per week, and dummy variables for gender, Asian, black, latino and Middle Eastern. We ran these regressions separately for Spectator, Anonymous Stakeholder and Known Stakeholder decisions and for US, Japan and pooled US/Japan subjects resulting in nine estimations. The entitlement continues to be positive and significant in every equation, but almost nothing else is significant: age, earnings, parents' income, work hours and race are never significant at conventional levels. Gender is significant (P-value=.033) in only one case: Japanese women give the X subject somewhat more in the Spectator treatment, but that is most likely apocryphal. Student expenditures are also insignificant, except in the Anonymous Stakeholder decisions, where expenditures are directly related to the fraction taken by US (P-value=.047), Japanese (P-value=.014) and pooled dictators (P-value=.004). This seems reasonable: "big spenders" take more for themselves when they are able to so do (i.e., are stakeholders) and can do so with impunity due to anonymity.

The Known Stakeholder decision offers the opportunity to explore whether dictators allocate differently based not just on their own race or gender but also on the race or gender of their recipients.¹⁵ To examine possible in-group/out-group race biases, dummy variables are added to the right hand side of equation (1) for each of the following dictator/recipient pairings: Asian/Asian, Asian/non-Asian, black/black, black/non-black, latino/latino, latino/non-latino and white/non-white, whereby white/white is the omitted category (there were no other racial groups represented in this decision). None of the race dummy variables is significant at conventional levels. Another possibility is that racial bias is not based on the in-group/out-group distinction but rather on the race of the recipient, e.g., it could be that non-blacks as a group are less generous towards blacks. To test this, we add dummies to equation (1) for the following dictator/recipient pairings: Asian/Asian, non-Asian, black/black, non-black/black, latino/latino, non-latino/latino and non-white/white, whereby white/white is the omitted category. Again, none of the coefficients on the race dummies is significant.

We can also examine whether knowing the gender of one's recipient significantly impacts allocations in the Known Stakeholder decision. On average, men take 55% of the pie and women 54%, roughly equal shares. But both groups take somewhat more for themselves when matched with men (56% for both male and female dictators) than when matched with women (53% for male dictators and 51% for female dictators). These are averages for the pooled sample, but we can analyze gender differences by nationality and controlling for entitlements. Table VIII reports the results of Tobit regressions of A subject (i.e., XA and YA) allocations on

¹⁵ Since the Japanese subjects were all Asian, the racially diverse pairings in the following analysis were entirely in the US sample. All of these findings hold, though, for both the US sample alone and the pooled US/Japan sample.

the entitlement and dummies for three dictator/recipient pairings: male with female, female with female and female with male, where male with male is the omitted category. The estimated intercept and entitlement slope coefficients and their significance are consistent with previous estimates of Known Stakeholder allocations, indicating as before a mixture of equality and equity. Interestingly, gender does not matter for Japanese dictators or in the regression for the pooled sample. American dictators of both genders, however, take less when paired with female counterparts, although this is only marginally significant in the case of male dictators. In the US, women matched with women take 17 percentage points less (relative to men matched with men). This pattern differs from the lesser generosity of women toward other women in Ben-Ner, Kong and Putterman (2004) but is consistent with the greater generosity toward women (especially by women) in Dufwenberg and Muren (2006) and Eckel and Grossman (2001).¹⁶ The absence of a gender effect in Japan might be related to the fact that Japanese Known Stakeholders are less selfish than their US counterparts, leaving little room for them to be even fairer to women.

	NATIONALITY OF DICTATOR				
Indep. Var.	US	Japan	Pooled US/Japan		
Intercept	0.50***	0.27***	0.31***		
-	(0.10)	(0.04)	(0.04)		
Entitlement	0.37***	0.51***	0.46***		
	(0.12)	(0.07)	(0.07)		
Male with female	-0.14*	0.01	-0.01		
	(0.07)	(0.04)	(0.03)		
Female with female	-0.17**	-0.03	-0.03		
	(0.07)	(0.09)	(0.03)		
Female with male	-0.11	-0.05	-0.02		
	(0.07)	(0.03)	(0.03)		
Adjusted R^2	0.24	0.49	0.33		
No. of obs.	44	47	91		
Female with male Adjusted <i>R</i> ² No. of obs.	-0.11 (0.07) 0.24 44	-0.05 (0.03) 0.49 47	-0.02 (0.03) 0.33 91		

 TABLE VIII

 TOBIT REGRESSIONS FOR KNOWN STAKEHOLDER ON GENDER

Notes: The omitted category is male dictator matched with male recipient. Significant at *10%, **5%, ***1%.

¹⁶ Consistent with conclusions in Dufwenberg and Muren (2006) and Holm and Engseld (2005), Olof Johansson-Stenman has suggested to us that this might be a case of the experiment impacting the external world rather than vice versa: subjects might be compensating in the experiment for perceived disadvantages of women outside the lab.

V. Discussion and Conclusions

The results of this study support two fairness rules, equity and equality, the relative importance of which depends systematically on personal factors. Impersonal third parties prefer equity, i.e., allocating rewards in proportion to contributions individuals control. Fairness preferences shift away from equity and toward equality as personal factors are added: stakeholders equalize more than spectators and again more when known by and to their counterparts. Consistent with a trade-off between equity and equality, Known Stakeholders weigh equity more heavily when stakes are higher; indeed, controlling for the size of stakes, dictators in this condition completely equalize allocations. The main findings are robust across a wide range of personal characteristics, including nationality, culture, race and gender.

We believe the results of this experiment are interesting, not only for the variables that are significantly related to allocation behavior, but also for the type and large number of those that are not. Fairness preferences appear to be unrelated to a considerable array of variables, including age, income, work hours, race, cultural orientation, and, usually, gender and expenditures. The only systematic difference across countries is the stricter compliance with fairness rules, whether equity or equality, by Japanese subjects contrasted with evidence of self interest on the part of anonymous American stakeholders, both individually and in groups.

By nesting experimental conditions and examining a wide range of variables, this study produces results about the importance of structural parameters that auger well for efforts to construct behavioral theories of social preferences. The findings are consistent with shared moral preferences, indeed, even with a shared sense of the relative importance of moral rules when multiple rules conflict, as with equity and equality. Although most non-ethics variables are not significant, the exceptions relate not to differences in values but in the willingness to act on them, as with the greater self-interest of anonymous big spenders and anonymous US stakeholders. In addition, individuals who belonged to more selfish groups became themselves more selfish in subsequent decisions, a result that can be seen as relevant to calls for corporate accountability and the development of an ethical business culture as means to discourage individual impropriety. Of course, these findings can only be validated with replication, but we believe that the results of many existing studies are consistent with the view that there are common moral preferences but individual heterogeneity in the interpretation of the rules and in the trade-off between rules and self-interest. For example, the difference between spectator

equity and the stakeholder shift toward equality is also observed in Gächter and Riedl (2006) and Konow (2000). And many studies have found considerable individual heterogeneity in weights placed on self-interest versus fairness, e.g., Fehr and Schmidt (1999), and between different rules of fairness, e.g., Cappelen, Hole, Sørensen and Tungodden (2007) and Charness and Rabin (2002). Regarding national differences, a greater general tendency for Japanese to comply with moral rules relative to Americans could also explain the fact that the former punish low contributions to public goods more vigorously, e.g., Cason, Saijo and Yamato (2002). Indeed, such differences in the relative salience of moral rules or in the willingness to act on them can reasonably explain a wide range of differences in international bargaining experiments, e.g., Henrich, et al. (2001).

Understanding what triggers equity and equality preferences is also potentially important for empirically based normative theory and for more effective economic policy. The accountability principle does not, as mentioned earlier, necessarily imply large inequality, since productivity differences should only be compensated to the extent they result from variables individuals control, such as effort or choices, and not from those they do not, such as brute luck or birth. There was widespread agreement that subjects were responsible for their performance on the task in this experiment, so the fairness of proportional allocations was not seriously in dispute. As Alesina and Angeletos (2005) point out, however, perceptions of responsibility for contributions vary systematically and have important implications for the perceived fairness of market outcomes and for redistribution policies (which, in turn, can affect the fairness of outcomes). Thus, it seems likely that observed patterns of support for redistribution both within and across countries reflect mostly differences in perceived responsibility, rather than differences in support for equily or equality per se.

This paper concludes with some brief thoughts about how the typology it proposes for fairness rules might be generalized to moral preferences as a whole. We relate this to the rapidly growing literature on identity and norms and suggest how this conceptual distinction might help provide explanations for other puzzles. We distinguish two types of moral preferences: *morals* and *mores*. Morals refer to the moral preferences of an impartial or impersonal party, and we call their rules *moral principles*. In the context of this experiment, equity is a moral principle. Mores, by contrast, are the moral preferences that are distinct to personal relationships, and their rules are referred to as *moral norms*. In our study, equality is a moral norm.

This choice of terminology reflects our attempt to stay close to common usage (although, of course, these terms have often been used interchangeably in the social preferences literature). Whereas morals usually connote general views of right and wrong, mores are often thought of as the "morally binding customs of a particular group," which implies a certain specificity to social context. Brekke, Kverndokk and Nyborg (2003) provide a theory and empirical evidence of how moral norms can adjust endogenously to context, sometimes with unexpected policy implications. Moral norms (and social norms generally) are also often associated with real or potential sanctions. Testing implications of the model of identity that Akerlof and Kranton (2000) introduced into economics, McLeish and Oxoby (2007) report experimental evidence of both cooperation and punishment consistent with material sanctions. Indeed, Ellingsen and Johannesson (2008) find that even the anticipation of anonymous verbal feedback increases altruistic behavior. These and other studies demonstrate the importance of social norm enforcement. But we argue this is not the defining feature of norms. People also punish and reward others for self-interested reasons, as in repeated games, and Fehr and Fischbacher (2004) demonstrate that even third parties are willing to incur costs to punish others. Moreover, the current study shows that people enforce norms even when sanctions are ruled out and only unilateral and anonymous decisions are allowed. Instead, we believe that personal factors are the critical feature that distinguishes norms from principles.

Naturally, morals and mores need not conflict, and the typical absence of such a distinction reflects the implicit assumption that they usually do not. Nevertheless, one can easily find cases of moral principles diverging from moral norms, for instance, in comparisons of third party allocations with the decisions of participants in social dilemmas, such as public good games. Third parties care about entitlements, as observed in the current study, as well as unequal endowments, as witnessed in Fehr and Fischbacher (2004). Yet contributions to public goods are often characterized by attempts to match contributions irrespective of endowment heterogeneity, as Buckley and Croson (2006) show, and of whether endowments are earned, as Cherry, Kroll and Shogren (2005) find. These results imply that a simple norm of equality sometimes replaces more complicated principles as means for solving coordination problems among stakeholders, consistent with the findings of Crawford et al. (2008).¹⁷

 $^{^{17}}$ That study employs a level-*k* model to account for variation in the effects of focal points. The authors suggest that certain results inconsistent with that model be explained by team reasoning.

Future research could explore other factors that might impact the trade-off between equity and equality, such as the content of communication between agents or the nature of employer-employee relationships. One might also examine possible differences between principles and norms with other types of moral preferences. For example, Croson and Konow (2009) find such differences in reciprocal preferences: stakeholders punish more and reward less than spectators. Other work could explore how moral norms evolve and how and why they sometimes differ from moral principles, including questions of how they relate to the principles of efficiency or need and potentially affect cooperation in social dilemmas.

A simple alternative (though not mutually exclusive) explanation for those inconsistencies is that higher types believe that the non-strategic type prefers equality. Level-*k* thinking anchored on this type provides an additional reason to expect the equality norm to emerge in such situations.

REFERENCES

- Aristotle (1925). *Ethica Nicomachea*. Translated by W. D. Ross. London: Oxford University Press.
- Akerlof, George A., and Kranton, Rachel (2000). "Economics and Identity," *Quarterly Journal* of Economics 115(3), 715-753.
- Alesina, Alberto, and Angeletos, George-Marios (2005). "Fairness and Redistribution," *The American Economic Review* 95(4), 960-980.
- Ball, Sheryl, Eckel, Catherine, Grossman, Philip J., and Zame, William (2001). "Status in Markets," *Quarterly Journal of Economics* 116(1), 161-188.
- Ben-Ner, Avner, Kong, Fanmin, and Putterman, Louis (2004). "Share and Share Alike? Gender-Pairing, Personality, and Cognitive Ability as Determinants of Giving," *Journal of Economic Psychology* 25, 581-589.
- Bohnet, Iris, and Frey, Bruno S. (1999). "The Sound of Silence in Prisoner's Dilemma and Dictator Games," *Journal of Economic Behavior and Organization* 38, 43-57.
- Bolton, Gary, Katok, Elena, and Zwick, Rami (1998). "Dictator Game Giving: Rules of Fairness versus Acts of Kindness," *International Journal of Game Theory* 27(2), 269-299.
- Bolton, Gary E., and Ockenfels, Axel (2000). "ERC: A Theory of Equity, Reciprocity, and Competition," *American Economic Review* 90(1), 166-193.
- Brandts, Jordi, Saijo, Tatsuyoshi, and Schram, Arthur (2004) "How universal is behavior? A four country comparison of spite and cooperation in voluntary contribution mechanisms," *Public Choice* 119 (3-4), 381-424.
- Brekke, Kjell Arne, Kverndokk, Snorre, and Nyborg, Karine (2003). "An Economic Model of Moral Motivation," *Journal of Public Economics* 87, 1967-1983.
- Buchan, Nancy, and Croson, Rachel (2004). "The Boundaries of Trust: Own and Others' Actions in the US and China," *Journal of Economic Behavior and Organization* 55, 485-504.
- Buckley, Edward, and Croson, Rachel (2006). "Income and Wealth Heterogeneity in the Voluntary Provision of Linear Public Goods," *Journal of Public Economics* 90, 935-955.
- Camerer, Colin (2003). *Behavioral Game Theory: Experiments in Strategic Interaction*, Princeton: Princeton University Press.
- Cappelen, Alexander, Hole, Astri, Sørensen, Erik, and Tungodden, Bertil (2007). "The pluralism of fairness ideals: An experimental approach," *American Economic Review* 97 (3), 818-827.
- Cason, Timothy, and Mui, Vai-Lam (1997). "A Laboratory Study of Group Polarization in the Team Dictator Game," *The Economic Journal* 107, 1465-1483.
- Cason, Timothy, Saijo, Tatsuyoshi, and Yamato, Takehiko (2002). "Voluntary Participation and Spite in Public Good Provision Experiments: An International Comparison," *Experimental Economics* 5(2), 133-153.
- Charness, Gary, and Gneezy, Uri (2008). "What's in a Name? Anonymity and Social Distance in Dictator and Ultimatum Games," *Journal of Economic Behavior and Organization* 68(1), 29-35.
- Charness, Gary, and Rabin, Matthew (2002). "Understanding Social Preferences with Simple Tests." *Quarterly Journal of Economics* 117(3), 817-869.
- Cherry, Todd L., Kroll, Stephan, and Shogren, Jason F. (2005). "The Impact of Endowment Heterogeneity and Origin on Public Good Contributions: Evidence from the Lab," *Journal of Economic Behavior and Organization* 57(3), 357-365.

- Corlett, J. Angelo (2003) "Making More Sense of Retributivism: Desert as Responsibility and Proportionality," *Philosophy* 78(2), 279-287.
- Crawford, Vincent, Gneezy, Uri, and Rottenstreich, Yuval, (2008). "The Power of Focal Points Is Limited: Minute Payoff Asymmetry May Yield Large Coordination Failures," *American Economic Review* 98(4), 1443-1458.
- Croson, Rachel and Uri Gneezy (forthcoming). "Gender Differences in Preferences," *Journal of Economic Literature*.
- Croson, Rachel and James Konow (2009). "Social Preferences and Moral Biases," *Journal of Economic Behavior and Organization*, 69(3), 201-212.
- D'Exelle, Ben, and Arno Riedl (2008). "Directed Generosity in Social and Economic Networks," manuscript.
- Dufwenberg, Martin, and Astri Muren (2006). "Generosity, Anonymity, Gender," *Journal of Economic Behavior and Organization* 61,42-49.
- Eckel, Catherine C., and Philip J. Grossman (2001). "Chivalry and Solidarity in Ultimatum Games," *Economic Inquiry* 39(2), 171-188.
- Ellingsen, Tore, and Johannesson, Magnus (2008). "Anticipated Verbal Feedback Induces Altruistic Behavior," *Evolution and Human Behavior* 29(2), 100-105.
- Fehr, Ernst and Fischbacher, Urs, (2004). "Third Party Punishment and Social Norms," *Evolution and Human Behavior*, 25, 63-87.
- Fehr, Ernst, and Schmidt, Klaus M. (1999). "A Theory of Fairness, Competition and Cooperation." *Quarterly Journal of Economics* 114(3), 817-868.
- Fong, Christina, and Luttmer, Erzo (2009). "What Determines Giving to Hurricane Katrina Victims? Experimental Evidence on Racial Group Loyalty," *American Economic Journal: Applied Economics*, 1(2), 64-87.
- Frank, Robert H. (2004). *What Price the Moral High Ground?* Princeton and Oxford: Princeton University Press.
- Frohlich, Norman, Oppenheimer, Joe, and Kurki, Anja (2004). "Modeling other-regarding preferences and an experimental test," *Public Choice* 119(1-2), 91-117.
- Frohlich, Norman, Oppenheimer, Joe, and Moore, Bernard (2001). "Some doubts about measuring self-interest using dictator games: the cost of anonymity," *Journal of Economic Behavior and Organization* 46, 250–271.
- Gächter, Simon, and Riedl, Arno (2005). "Moral Property Rights in Bargaining with Infeasible Claims," *Management Science*, 51(2), 249-263.
- Gächter, Simon, and Riedl, Arno (2006). "Dividing Justly in Bargaining Problems," *Social Choice and Welfare*, 27, 571-594.
- Greenberg, Jerald (2001). "Studying Organizational Justice Cross-Culturally: Fundamental Challenges," *The International Journal of Conflict Management* 12, 365-375.
- Güth, Werner (1988). "On the Behavioral Approach to Distributive Justice: A Theoretical and Experimental Investigation." In *Aspects of Distributive Justice*, edited by Shlomo Maital. New York: New York University Press, 703-17.
- Güth, Werner (1994). "Distributive Justice: A Behavioral Theory and Empirical Evidence." In *Essays on Economic Psychology*, edited by H. Brandstätter and Werner Güth, 153-75.
- Henrich, Joseph, Boyd, Robert, Bowles, Samuel, Camerer, Colin, Fehr, Ernst, Gintis, Herbert and McElreath, Richard (2001). "In Search of Homo Economicus: Behavioral Experiments in 15 Small-Scale Societies," *American Economic Review* 91(2), 73-78.

- Hoffman, Elizabeth, McCabe, Kevin, Shachat, Keith, and Smith, Vernon (1994). "Preferences, Property Rights, and Anonymity in Bargaining Games," *Games and Economic Behavior* 7, 346-80.
- Hoffman, Elizabeth, Kevin McCabe, and Vernon L. Smith (1996). "Social Distance and Other-Regarding Behavior in Dictator Games," *American Economic Review* 86(3), 653-660.
- Holm, Håkan J., and Danielson, Anders (2005). "Tropic Trust versus Nordic Trust: Experimental Evidence from Tanzania and Sweden," *The Economic Journal*, 115(April), 505-532.
- Holm, Håkan J., and Engseld, Peter (2005). "Choosing Bargaining Partners An Experimental Study on the Impact of Information about Income, Status and Gender," *Experimental Economics*, 8(3), 183-216.
- Johansson-Stenman, Olof, Mahmud, Minhaj, and Martinsson, Peter (2005). "Does Stake Size Matter in Trust Games?," *Economics Letters* 88, 365-369.
- Kashima, Yoshihisa, Siegel, Michael, Tanaka, Kenichiro, and Isaka, Hiroko (1988). "Universalism in Lay Conceptions of Distributive Justice: A Cross-Cultural Examination," *International Journal of Psychology* 23, 51-64.
- Kluegel, James R., and Smith, Eliot R. (1986). Beliefs About Inequality: Americans' Views of What is and What Ought to Be, New York: Aldine De Gruyter.
- Konow, James (2000). "Fair Shares: Accountability and Cognitive Dissonance in Allocation Decisions," *American Economic Review* 90(4), 1072-92.
- Konow, James (2003). "Which is the Fairest One of All?: A Positive Analysis of Justice Theories," *Journal of Economic Literature*, 41(4), 1186-1237.
- Kroll, Stephan, Cherry, Todd L., and Shogren, Jason F. (2007). "The Impact of Endowment Heterogeneity and Origin on Contributions in Best-Shot Public Good Games," *Experimental Economics* (2), pp. 133-153.
- Lemieux, Thomas, MacLeod, W. Bentley, and Parent, Daniel (2009). "Performance Pay and Wage Inequality." *Quarterly Journal of Economics* 124(1), 1-49.
- Levine, David K. (1998). "Modeling Altruism and Spitefulness in Experiments," *Review of Economic Dynamics*, 1(3), 593-622.
- Levitt, Stephen D., and List, John A. (2007). "What Do Laboratory Experiments Measuring Social Preferences Reveal about the Real World?," *Journal of Economic Perspectives*, 21(2), 153-174.
- List, John A. (2007). "On the Interpretation of Giving in Dictator Games," *Journal of Political Economy* 115(3), 482-493.
- Luhan, Wolfgang J., Kocher, Martin G., and Sutter, Matthias (2009). "Group Polarization in the Team Dictator Game Reconsidered," *Experimental Economics* 12(1), pp. 26-41.
- Mann, Leon, Radford, Mark, and Kanagawa, Chie (1985). "Cross-Cultural Differences in Children's Use of Decision Rules: A Comparison between Japan and Australia," *Journal of Personality and Social Psychology* 49(6), 1557-1564.
- Mattila, Anna S., and Patterson, Paul G. (2004). "Service Recovery and Fairness Perceptions in Collectivist and Individualist Contexts," *Journal of Service Research* 6, 336-346.
- McLeish, Kendra N., and Oxoby, Robert J. (2007). "Identity, Cooperation, and Punishment," University of Calgary; University of Calgary and IZA, January.
- Mill, John Stuart. (1861) [1979]. *Utilitarianism*. Indianapolis, Cambridge, Massachusetts: Hackett Publishing Company, Inc.
- Ohtake, Fumio, and Saito, Makoto (1998). "Population Aging and Consumption Inequality in Japan," *Review of Income and Wealth* 44, 361-381.

Okada, Akira and Arno Riedl (1999). "When Culture Does Not Matter: Experimental Evidence from Coalition Formation Ultimatum Games in Austria and Japan," Tilbergen Institute discussion paper.

Rawls, J. (1971): A Theory of Justice, Cambridge: Belknap Press of Harvard University Press.

- Rege, Mari, and Telle, Kjetil (2004). "The Impact of Social Approval and Framing on Cooperation in Public Good Situations," *Journal of Public Economics* 88(7-8), 1625-1644.
- Roth, Alvin, Prasnikar, Vesna, Okuno-Fujiwara, Masahiro, and Zamir, Shmuel (1991).
 "Bargaining and Market Behavior in Jerusalem, Ljubljana, Pittsburgh, and Tokyo: An Experimental Study," *American Economic Review* 81(5), 1068-1095.
- Selten, Reinhard (1978). "The Equity Principle in Economic Behavior." In *Decision Theory and Social Ethics, Issues in Social Choice*, edited by H. Gottinger and W. Leinfellner. Dordrecht: Reifel Publishing, 289-301.
- Singelis, Theodore M., Triandis, Harry C., Bhawuk, Dharm P. S., and Gelfand, Michele J. (1995). "Horizontal and Vertical Dimensions of Individualism and Collectivism: A Theoretical and Measurement Refinement." *Cross-Cultural Research*, 29(3), 240-275.
- Tachibanaki, Toshiaki (2005). Confronting Income Inequality in Japan: A Comparative Analysis of Causes, Consequences, and Reform, Cambridge: The MIT Press.
- Walzer, Michael (1983). Spheres of Justice: A Defense of Pluralism and Equality. New York: Basic Books, Inc.
- Xiao, Erte, and Houser, Daniel (2007). "Emotion Expression and Fairness in Economic Exchange." http://repec.ices-gmu.org/RePEc/pdf/1003.pdf.